Author's response to reviews

Title: Web-based Screening and Brief Intervention for Poly-drug use among Teenagers: Study protocol of a multicentre two-arm randomized controlled trial

Authors:

Nicolas Arnaud (n.arnaud@uke.de)
Sonja Broening (s.broening@uke.de)
Magdalena Drechsel (m.drechsel@uke.de)
Rainer Thomasius (r.thomasius@uke.de)
Christiane Baldus (c.baldus@uke.de)

Version: 2 Date: 31 August 2012

Author’s response to reviews: see over
Cover letter

Dear Editors at BMC Public Health,

Thank you to consider our manuscript “Web-based Screening and Brief Intervention for Poly-drug use among Teenagers: Study protocol of a multicentre two-arm randomized controlled trial” for publication in BMC Public Health.

You asked us to include and name all ethics committees involved in our study. The revised version of the manuscript includes mentioning of all ethical committees involved in the study. They are explicitly mentioned on page 7 in the last sentence under the header "Study Flow".

This sentence reads as follows now: "Approval for the design and data collection was already obtained from the responsible authorities in each participating country, which are the Ethical Committee of the Chamber of Physicians Hamburg (Germany), Prague Psychiatric Centre (Czech Republic), University Hospital of Antwerp and the University of Antwerp (Belgium), and the Regional Ethics Board in Stockholm (Sweden)."

Sincerely,

Dr. Nicolas Arnaud and colleagues